Author’s response to reviews

Title: Developing Optimal Search Strategies for Detecting Clinically Sound Prognostic Studies in MEDLINE: An Analytic Survey

Authors:

Nancy L Wilczynski (wilczyn@mcmaster.ca)
Prof R. BRIAN Haynes (bhaynes@mcmaster.ca)

Version: 3 Date: 11 May 2004

Dear Editor,

The comments from the reviewer are as follows: I find the authors revisions and comments to the manuscript acceptable. My initial comments are answered and the changes made in the manuscript seem adequate.

Thus, no further revisions were made to the manuscript based on reviewer comments.

Yours truly,
Nancy Wilczynski